Author’s response to reviews

Title: Prevalence and correlates of Diabetes mellitus in Malawi: Population-based national NCD STEPS Survey

Authors:

Kelias P Msyamboza (msyambozak@who.int)
Chimwemwe Mvula (cmvula@yahoo.com)
Damson Kathyola (dkathyola@gmail.com)

Version: 3 Date: 13 April 2014

Author’s response to reviews: see over
The Editor
BMC Endocrine Disorders

Dear Sir/Madam,

COVERING LETTER:

MS: 8882499531156550 - Prevalence and correlates of Diabetes mellitus in Malawi: Population-based national NCD STEPS Survey

On behalf of my fellow authors, I hereby submit the above named revised manuscript.

We are sincerely grateful for comments, suggestions and guidance that we received from the editor and the two reviewers. We have worked on all the comments and incorporated all the suggestions.

Attached are:
- Comment-by-comment responses
- Revised manuscript with corrections highlighted or in tracks changes
- Clean revised manuscript.

Once again, thank you for your time and valuable comments, suggestions and guidance. The manuscript is now in better form.

Yours faithfully

Dr Kelias Msyamboza
(Corresponding author)